Author's response to reviews

Title: Lipid-soluble smoke particles damage endothelial cells and reduce endothelium-dependent dilatation in rat and man

Authors:
Jin-Yan Zhang (zhangjinyan1978@163.com)
Yong-Xiao Cao (yxxy@xjtu.edu.cn)
Cang-Bao Xu (Cang-Bao.Xu@med.lu.se)
Lars Edvinsson (Lars.Edvinsson@med.lu.se)

Version: 5 Date: 20 December 2005

Author’s response to reviews:

Dear Editor,

Thank you very much for your kind letter and requesting us to do formatting changes of our manuscript. We have done the changes according to your instruction and comment. Below is a point-by-point description:

1. Title, abstract and author list are exactly in our manuscript and in the submission system.

2. Keyword section is now removed from our manuscript as suggested.

3. Figure 1, 2, 3 and 4 are now removed from the image files as suggested.

4. The typo in the background section has now been corrected as suggested.

Yours sincerely

Cang-Bao Xu, MD, Ph.D
Associate Professor in Medicine
Lund University